Author's response to reviews

Title: Early postoperative cognitive dysfunction and postoperative delirium after anaesthesia with various hypnotics: a double blind, prospective, randomized, controlled clinical trial - The PINOCCHIO trial -

Authors:

Federico Bilotta (bilotta@tiscali.it)
Andrea Doronzio (a.doronzio@libero.it)
Elisabetta Stazi (elisabetta_stazi@yahoo.it)
Luca Titi (lucatiti@hotmail.it)
Carmelo Violi (c_viol85@yahoo.it)
Ivan Orlando Zeppa (jakie84@libero.it)
Giovanni Rosa (giov.rosa@libero.it)
Francesca Paola Paoloni (francesca_paoloni@yahoo.it)
Vincenzo Fodale (vfodale@unime.it)
Letterio Santamaria (Letterio.Santamaria@unime.it)
Rita Melotti (rita.melotti@alice.it)
Gianfranco Di Nino (Gianfranco.dinino@unibo.it)
Cesare Vittori (cesare.vittori@tin.it)
Ettore Zei (E_zei@yahoo.it)
Elisabeth Apollonia Abramowicz (Elisabeth.abramowicz@gmail.com)
Allison Spinelli (aspinelli@montefiore.org)
Eugenia Ayrian (eugenia.ayrian@gmail.com)
Vladimir Zelman (vzelman@hsc.usc.edu)

Version: 3 Date: 28 August 2010

Author's response to reviews:

Dear Editor,

find enclosed the revised version of the manuscript. We have provided more details about ethical committee and we have received a attendance of 2 colleagues from the University of South California. These authors are Eugenia Ayrian (eugenia.ayrian@gmail.com) and Vladimir Zelman (vzelman@hsc.usc.edu). All authors agree on the revised list. Furthermore all authors have studied and approved the final submitted version of protocol.

thank you for your attention

Sincerely

Federico Bilotta MD, PhD